Accelerator Status Report

Run Schedule: Aug 21, 2009 to Dec 23, 2009
Nov. 4, 2009

Injector

· The injector needs optimized to run high current (over 100uA).

· (11/2)Injector was set to 67.5 MeV for the high energy run.

· (11/2) The GUN2 photo cathode was activated.
· (10/30) Iocin3 crashed. After the crash recovery the GUN HV did not come back gracefully. 

· Cavity 0L04-2 was by-passed and recovered for the high energy run
· (10/28) Recovery from the power outage (10/27, Tuesday) went well after RF was up

· After the Tuesday’s power outage, while troubleshooting the GUN HV problems the injector group found a problem with the splice between the GUN HV cable going into the GUN ceramic and the cable coming from the Ross probe tank upstairs in the service building. They believed that the problem at the splice is that there was an arc to ground. The problem was fixed. This may be the source of the injector instability.
Accelerator   Energy: 6.068 GeV
· (11/2) the accelerator was scaled from 5.764 to 6.068GeV: good optics.
· Isuuses with file downloads

· FSD mask issue with BSY. Hall A line vacuum valve is not fuuly protected.
· MO adjusted after the energy change, which resulted in a lot of path length adjustment.

· EES recovered some RF to support the high energy run.

· (10/28) Recovery from the power outage (10/27, Tuesday) went well once RF was up. Sine Hall A was down and Hall was at pass 1 and Hall C pass2, only the first 2-pass beam was checked. 

Hall A---PVDIS; pass 5; energy 6.068 GeV
· Hall A has changed over to PVDIS.

· (10/28) Hall A decided to end its data taking after the power outage.

· Hall A has been having problems with its target and magnet. The hall soon will change to a new experiment.

· Hall A Moller MST download: if the load button is not active push the override button to activate it.
Hall B ---Eg6; pass 5; energy 6.068 GeV
· Always send beam to the Tagger dump when tuning the hall line.

· Hall B beam was not stable a couple of times: BOM trips, current fluctuations, railed attenuator. It was due to orbit drift in the injector and 2T line. Steering orbit back resolved the problems.
Hall C---HKS; pass 2; energy 2.344 GeV
· HKS ended successfully on Nov. 2.

· Hall C will be down for 7months for Qweak installation.

· Hall C beam has been fine after the power outage.
